**Author’s response to reviews**

**Title:** Predictive value of aspartate aminotransferase-to-platelet ratio index for parenteral nutrition-associated cholestasis in infants with extremely low birth weight

**Authors:**

Ji Hwang (H00165@paik.ac.kr)

Mi Lim Chung (forevery52@naver.com)

**Version:** 1  **Date:** 25 Feb 2019

**Author’s response to reviews:**

Dear Editor:

I, along with my coauthors, would like to re-submit the attached manuscript entitled “Predictive value of aspartate aminotransferase-to-platelet ratio index for parenteral nutrition-associated cholestasis in infants with extremely low birth weight” as an original article. The manuscript Number is BPED-D-18-01055.

The manuscript has been carefully rechecked and appropriate changes have been made in accordance with the reviewers’ comments. The responses to the reviewers’ comments have also been prepared.

All the authors have approved the manuscript and declare no competing interests associated with this manuscript.

We hope that the revised manuscript is now suitable for publication in your journal.

I look forward to your reply.

Sincerely,

Mi Lim Chung, MD.